Alternative community-based models of care for young people with anorexia nervosa: the CostED national surveillance study

Sarah Byford,^{1*} Hristina Petkova,¹ Ruth Stuart,¹ Dasha Nicholls,² Mima Simic,³ Tamsin Ford,⁴ Geraldine Macdonald,⁵ Simon Gowers,⁶ Sarah Roberts,¹ Barbara Barrett,¹ Jonathan Kelly,⁷ Grace Kelly,⁸ Nuala Livingstone,⁸ Kandarp Joshi,^{9,10} Helen Smith¹¹ and Ivan Eisler^{1,3}

¹Health Service and Population Research Department, Institute of Psychiatry, Psychology & Neuroscience, King's College London, London, UK

Declared competing interests of authors: Tamsin Ford is chairperson of the Child and Adolescent Psychiatry Surveillance Service that was used to run part of the study, which is an unpaid position (other than travel expenses). Kandarp Joshi reports grants from Sunovion Pharmaceuticals, Inc., outside the submitted work. He was principal investigator for the Aberdeen site for a Sunovion-sponsored multisite trial on the efficacy and safety of lurasidone in paediatric schizophrenia. Jonathan Kelly reports that Beat has contracts with some NHS trusts and clinical commissioning groups that provide and commission, respectively, community eating disorders services for children and young people. In these contracts, Beat works with the local services to deliver awareness-raising training for professionals and, in some cases, to offer peer coaching for carers. The funding Beat receives in return for its role in this project is not 'on my behalf'. Beat is a co-applicant on another research project called 'TRIANGLE' (short title), which is also funded by the National Institute for Health Research. The funding Beat receives in return for its role in this project is not 'on my behalf'. Beat is a co-applicant on another research project called 'Mind the Gap' (short title), which is funded by the Welsh Assembly Government and is investigating referral pathways from identification to specialist treatment for people with eating disorders (of all ages) in Wales.

²Department of Medicine, Imperial College London, London, UK

³South London and Maudsley NHS Foundation Trust, London, UK

⁴Institute of Health Research, Department of Psychological Sciences, University of Exeter Medical School, University of Exeter, Exeter, UK

⁵School for Policy Studies, University of Bristol, Bristol, UK

⁶University of Liverpool, Liverpool, UK

⁷Beat, Norwich, UK

⁸School of Social Sciences, Education and Social Work, Queen's University Belfast, Belfast, UK

⁹Child and Adolescent Mental Health Service, Royal Cornhill Hospital, NHS Grampian, Aberdeen, UK

¹⁰Mental Health Division, University of Aberdeen, Aberdeen, UK

¹¹NHS Greater Glasgow and Clyde, Glasgow, UK

^{*}Corresponding author s.byford@kcl.ac.uk

Published October 2019 DOI: 10.3310/hsdr07370

Plain English summary

The CostED national surveillance study

Health Services and Delivery Research 2019; Vol. 7: No. 37

DOI: 10.3310/hsdr07370

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Previous research suggests that providing specialist eating disorders services for young people with anorexia nervosa could improve their outcomes and save money for the NHS by reducing the number of young people who are admitted to hospital and the length of time young people spend in hospital if they are admitted.

The aim of this study was to calculate the cost of supporting young people with anorexia nervosa who are being treated in two different types of community services: specialist eating disorders services and general child and adolescent mental health services (CAMHS). The study also investigated whether or not specialist services are better value for money than CAMHS and would save the NHS money.

The study identified young people with anorexia nervosa by contacting all child and adolescent psychiatrists in the UK and Republic of Ireland once per month for 8 months and asking them to inform the study if they had a new case of anorexia nervosa.

These psychiatrists gave us information on 298 young people (aged 8–17 years) who had a diagnosis of anorexia nervosa for the first time. These psychiatrists were asked to provide information on how these young people were doing when they were first diagnosed and then 6 months and 1 year later. The study also asked what health services the young people used during that year.

Young people in specialist services were found to be more severely ill than those in CAMHS when they were first diagnosed. Despite this, care for young people in specialist services cost about the same as care for those who were diagnosed in general CAMHS, and their outcomes after 1 year were also similar. This work showed that specialist services may be better value for money than general CAMHS, but it did not show that providing more specialist services would save money for the NHS. Decisions about how to organise services could, therefore, take other factors into consideration, such as the preferences of patients and carers.

Health Services and Delivery Research

ISSN 2050-4349 (Print)

ISSN 2050-4357 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HS&DR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health Services and Delivery Research journal

Reports are published in *Health Services and Delivery Research* (HS&DR) if (1) they have resulted from work for the HS&DR programme or programmes which preceded the HS&DR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

HS&DR programme

The HS&DR programme funds research to produce evidence to impact on the quality, accessibility and organisation of health and social care services. This includes evaluations of how the NHS and social care might improve delivery of services.

For more information about the HS&DR programme please visit the website at https://www.nihr.ac.uk/explore-nihr/funding-programmes/health-services-and-delivery-research.htm

This report

The research reported in this issue of the journal was funded by the HS&DR programme or one of its preceding programmes as project number 11/1023/17. The contractual start date was in June 2013. The final report began editorial review in February 2018 and was accepted for publication in August 2018. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HS&DR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HS&DR programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Byford et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Senior Clinical Researcher, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk